Physics 215A–Winter ’00

Instructor: M. Bander Office: FRH 3176 Phone: 824-5945 E-mail: mbander@uci.edu Web: You will find general information, old and new problem sets and old exams on http://www.physics.uci.edu/∼mbander .

Office hours: anytime Book: Sakurai, Modern Quantum Mechanics; Rev. Edition Hours: M,W 1-2:20 in FRH 3111 From time to time we will meet on Fridays, 9-10 in FRH 3111.

Homework: Assigned on Wednesdays, due the following Mondays. Midterm: February 9 Final: March 22, 1:30 to 3:30 Grading Policy: Final: 50% Midterm: 30% Homework: 20%